\begin{table}%t4
\caption{Objects excluded from the HEB/LEB sample.}
\label{tab:resultats_misc}
\small%\centerline
{
\begin{tabular}{lrrrrrrrrl}
\hline
\hline
Object & $F$(\hb) & \oiii~ 4959+5007&\ha  & $c$(\hb) & \oiii / \hb  & $I$(\hb) &
$L$(\hb) & Radius & Notes\\ 
 & 
$10^{-16}$&
$10^{-16}$&
$10^{-16}$&
$10^{-16}$&
&
$10^{-16}$&
$10^{26}$& 
pc&
\\
&\unit{W}/\unit[2]{m}  &
\unit{W}/\unit[2]{m}  &
\unit{W}/\unit[2]{m}  &
\unit{W}/\unit[2]{m}  &
&
\unit{W}/\unit[2]{m}  &
\unit{W} &
&
\\
\hline
\multicolumn{8}{l}{Planetary Nebula, LMC:}&\\
N~66 &   3.5 &  40.2 &  11.4 &  11.37 &  0.17 &   5.2 &    157 & 0.3 &(a) (b) \\
\multicolumn{8}{l}{Symbiotic star, SMC:}\\
N~73 &   0.7 &   0.1 &   7.2 &   0.20 &  1.89 &  51.5 &   2615 &0.3 &(c) \\
\multicolumn{8}{l}{Planetary Nebulae, SMC:}\\
N~1 &   2.5 &   9.1 &   9.8 &   3.64 &  0.45 &   7.1 &    361 &0.3 &(d) \\
N~9 &  18.0 &  69.3 &  51.6 &   3.84 &  0.00 &  18.1 &    917 &0.7&(e) \\
N~29 &   0.3 &   1.8 &   1.3 &   6.29 &  0.60 &   1.2 &     60 &0.3& (d) \\
N~47 &   3.9 &  16.3 &  12.4 &   4.18 &  0.15 &   5.5 &    280 & 0.3&(d) \\
N~61 &  10.0 &  22.4 &  35.7 &   2.25 &  0.31 &  20.6 &   1047 & 0.5& (f) \\
N~70 &   2.9 &  18.1 &  11.7 &   6.24 &  0.50 &   9.1 &    462 & 0.2&(d) \\
\hline
\end{tabular}}
\medskip
(a)~\citet{pena04};
(b)~\citet{morgan84};
(c)~\citet{walker83};
(d)~\citet{sanduleak78};
(e)~\citet{meyssonnier93};
(f)~\citet{jacoby02}.
\end{table}